Author's response to reviews

Title: Genetic association study of selected candidate genes (APOB, LPL, LEPTIN) and telomere length in obese and hypertensive individuals.

Authors:

Birajalaxmi Das (birajalaxmi@yahoo.co.in)
Nilambari Pawar (nilambari.pawar@gmail.com)
Divyalakshmi Saini (ndivya@barc.gov.in)
M Seshadri (msesh@barc.gov.in)

Version: 9 Date: 20 September 2009

Author's response to reviews:

To

Editor-in-Chief, 20th September 2009

BMC Medical Genetics,

Dear Sir/Madam,

With reference to the email dtd. 14th September 2009, we have made the formatting changes in the revised manuscript (MS: 9657831702395905) entitled, “genetic association study of selected candidate genes (ApoB, LPL, Leptin) and telomere length in obese and hypertensive individuals.” authored by Birajalaxmi Das, Nilambari Pawar, Divyalakshmi Saini and M. Seshadri and submitted to your esteemed journal . We have made the following changes as required by you as a proof copy:

Major revisions:

1. Changes in the author’s contributioins as per the requirement.

2. Broken link of reference 60 has been checked and corrected.

3. Figure titles and legends have been removed from the image and kept after the reference list in the manuscript file.

4. Changes in the layout of the Tables have been made as per the requirement.

5. All the authors have been listed in reference 40 and the term “et al” is removed.
Minor revisions:

1. The title of the manuscript has been changes to sentence case.

2. Running title/running head have been removed from the title page.

3. Email format has been given as per the requirement.

4. The “Materials and Methods” section has been renamed as “Methods” in the abstract and same has been changed in the submission system.

5. “Introduction” section has been renamed as “Background”.

6. All the highlights have been removed from the manuscript.

7. Table titles have been placed above the table.

8. Additional files have been removed.

9. Figure cropping has been done.

We sincerely thank BMC Biomedical Genetics for accepting our manuscript in the esteemed journal.

Thanking you,

Dr. Birajalaxmi Das.

birajalaxmi@yahoo.co.in